Author's response to reviews

Title: Clinical Research Evidence of Cupping Therapy in China: A Systematic Literature Review

Authors:

Huijuan Cao (huijuancao327@hotmail.com)
Mei Han (hanmeizoujin@hotmail.com)
Xum Li (tina000341@hotmail.com)
Shangjuan Dong (bener84.11.11@163.com)
Yongmei Shang (shangyongmei@tsinghua.org.cn)
Qian Wang (wangwangwangqian@yahoo.com.cn)
Shu Xu (tangtang817@163.com)
Jianping Liu (jianping_l@hotmail.com)

Version: 4 Date: 15 November 2010

Author's response to reviews: see over
Cover Letter

November 15, 2010

Dear Editor,

Thank you for accepting our manuscript, we did some revisions as marked in the text. Based on the query for our manuscript, we made the modifications. Below are the detailed responses to all the points.

Looking forward to hearing from you soon.

Best regards,

Huijuan Cao and author team

1. Responses to the Editor Query

EU1: References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].'

Reply: We revised it.

EU2: Authors' contributions: Please use initials instead of full name and indicate that all authors read and approved the final manuscript.

Reply: We revised it, also the names in Acknowledgement.

EU3: Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript. Please do not use ranges when listing figures. Please list all the figures, for example: Figures 5, 6, 7, 8 and 9.

Reply: We revised it.

EU4: Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Reply: We revised it.
EU5: Tables- please remove the visible vertical lines from your tables

Reply: We revised it.

EU6: Additional files: please remove these from the submission system.

Reply: We revised it.

EU7: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.

Reply: We revised it.

2. Other changes we made to the original manuscript

Revision1: We revised the error in figures.